Author’s response to reviews

Title: THE USE OF LATIN TERMINOLOGY IN MEDICAL CASE REPORTS: QUANTITATIVE, STRUCTURAL AND THEMATIC ANALYSIS

Authors:

Yuliia Lysanets (julian.rivage@gmail.com)

Olena Bieliaieva (e_lena69@rambler.ru)

Version: 1 Date: 29 Oct 2017

Author’s response to reviews:

Reviewer #1: The manuscript was re-structured as suggested by the Reviewer. The sections of METHODS, RESULTS and DISCUSSION were re-arranged.

The order of referencing was switched to the consecutive.

The authors included an explanation of the important role of Latin in the MCRs genre; we described how the search and selection of the material were done.

The timeline of selected publications, as well we the total number of papers and their types were specified.

Reviewer #2: The required points were addressed and answered:

1) The order of countries from which came the most common use of Latin terms was added in the DISCUSSION section

2) The three-stage plan of implementing the results of this research was provided in the CONCLUSIONS

3) The inclusion of Latin in MCRs module/extracurricular submodule was explained on the basis of existing training programmes in different countries.

The authors express their deep appreciation to the Reviewers for giving valuable insights and suggestions which helped to improve this manuscript significantly.